Author's response to reviews

Title: Disseminating quality improvement: study protocol for a large cluster randomized trial

Authors:

- David H. Gustafson (dhgustaf@wisc.edu)
- Andrew R. Quanbeck (andrew.quanbeck@chess.wisc.edu)
- James H. Ford II (jay.ford@chess.wisc.edu)
- Alice Pulvermacher (alice.pulvermacher@chess.wisc.edu)
- Michael T. French (mfrench@mail.as.miami.edu)
- K. J. McConnell (mcconnjo@ohsu.edu)
- Dennis McCarty (mccartyd@ohsu.edu)

Version: 7 Date: 14 April 2011

Author's response to reviews: see over
April 14, 2011

Dr. Martin Eccles
Dr. Brian Mittman
Editors-in-Chief
Implementation Science

Dear Doctors Eccles and Mittman:

My colleagues and I hereby submit the proofread version of our manuscript “Disseminating quality improvement: study protocol for a large cluster-randomized trial.” We have corrected typographical errors as instructed by the Independent Editorial Production Team.

Thank you for the opportunity to publish our work in your journal.

Cordially,

Andrew Quanbeck, M.S.
Associate Researcher
Center for Comprehensive Health Enhancement Systems Studies
4161B Mechanical Engineering Building
1513 University Avenue
Madison, WI 53706
Telephone: 608-890-1016
Fax: 608-890-1438
Email: andrew.quanbeck@chess.wisc.edu